Author's response to reviews

Title: Use of Real time Polymerase Chain Reaction for detection of M. tuberculosis, M. avium and M. kansasii from clinical specimen

Authors:

Arnold Bainomugisa (arnoldbaino@yahoo.co.uk)
Chris Muchwa (muchwachris@yahoo.com)
Joseph Akol (jpakol01@gmail.com)
Paul Mubiri (mubirip@gmail.com)
Sam Ogwang (sxo81@case.edu)
Enock Matovu (matovue@vetmed.mak.ac.ug)
Eddie Wampande (wamps@vetmed.mak.ac.ug)
Moses Joloba (mlj10@case.edu)

Version: 2 Date: 27 December 2014

Author's response to reviews:

Revised the main manuscript because the tables didn't look appear good. Figure 2 was also revised.